Author’s response to reviews

Title: Data linkage infrastructure for cross-jurisdictional health-related research in Australia.

Authors:

James H Boyd (J.Boyd@curtin.edu.au)
Anna M Ferrante (A.Ferrante@curtin.edu.au)
Christine M O'Keefe (Christine.O'Keefe@csiro.au)
Alfred J Bass (John.Bass@utas.edu.au)
Sean M Randall (Sean.Randall@curtin.edu.au)
James B Semmens (James.Semmens@curtin.edu.au)

Version: 2 Date: 27 August 2012

Author’s response to reviews:

This is a revised document containing the manuscript originally submitted as well as
- answers to all the competing interests questions
- a completed Authors Contributions
as requested.

The manuscript itself has not been changed.

We are happy to accept this as a correspondance article, rather than a research article.